Author’s response to reviews

Title: Condom use peer norms and self-efficacy as mediators between community engagement and condom use among Chinese men who have sex with men

Authors:
Haochu Li (hcli@alumni.cuhk.net; haochuli@med.unc.edu)
Li Xue (maydayxue1206@163.com)
Joseph Tucker (jdtucker@med.unc.edu)
Chongyi Wei (chongyiwei@hotmail.com)
Maya Durvasula (maya.durvasula@gmail.com)
Wenqi Hu (huwenqi_epi@163.com)
Dianmin Kang (dmkang66@163.com)
Meizhen Liao (liaomz161@126.com)
Weiming Tang (weimingtangscience@gmail.com)
Wei Ma (weima@sdu.edu.cn)

Version: 1 Date: 07 Nov 2016

Author’s response to reviews:

We have indicated as follow:

Ethics Approval and Consent to Participate

Ethical approval was granted by Institutional Review Board of the University of North Carolina, Chapel Hill; University of California, San Francisco; Shandong University, Shandong Center for Disease Control and Prevention, and Guangdong Provincial Center for Skin Diseases and STI Control. Written informed consent has been obtained from all participants.